Author's response to reviews

Title: Role of emmprin in endometrial cancer

Authors:

Keiichiro Nakamura (k-nakamu@cc.okayama-u.ac.jp)
Junichi Kodama (kodama@cc.okayama-u.ac.jp)
Atsushi Hongo (hongo1@cc.okayama-u.ac.jp)
Yuji Hiramatsu (kiki1063@cc.okayama-u.ac.jp)

Version: 4 Date: 2 May 2012

Author's response to reviews: see over
Dear BMC Cancer Editor

Thank you for your letter regarding our paper entitled “Role of emmprin in endometrial cancer” (BMC Cancer).

We have revised our manuscript according to the reviewer’s comments.

Q1. -Ethics - Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration, and any experimental research on animals must follow internationally recognized guidelines. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. Alternatively, please provide the name of the body which gave approval to use the data.

A1. Add the manuscript. This study was approved by the Institutional Review Board of Okayama University Hospital.

We hope that the revised manuscript will now be acceptable for publication in the BMC Cancer.

Sincerely yours,

Keiichiro Nakamura, M.D., Ph.D.
Department of Obstetrics and Gynecology,
Okayama University Graduate School of Medicine, Dentistry and Pharmaceutical Sciences,
2-5-1 Shikata-cho, Okayama 700-8558, Japan.
Phone: +81-86-235-7320
Fax: +81-86-235-9570
E-mail: k-nakamuc@cc.okayama-u.ac.jp